Author's response to reviews

Title: A phase-II trial of combined chemotherapy with S-1, oral leucovorin, and bevacizumab in heavily pre-treated patients with metastatic colorectal cancer

Authors:

Kazuhisa Yamaguchi (kazuhisa.yamaguchi@med.toho-u.ac.jp)
Hiroya Taniguchi (h.taniguchi@aichi-cc.jp)
Azusa Komori (a.komori@aichi-cc.jp)
Yukiya Narita (yukiya.narita@aichi-cc.jp)
Sohei Nitta (shntt2@gmail.com)
Motoo Nomura (mnomura@aichi-cc.jp)
Shigenori Kadowaki (skadowaki@aichi-cc.jp)
Daisuke Takahari (daisuke.takahari@jfc.or.jp)
Takashi Ura (tura@aichi-cc.jp)
Masashi Andoh (mandoh@aichi-cc.jp)
Kei Muro (kmuro@aichi-cc.jp)
Yoshinori Igarashi (igarashi@med.toho-u.ac.jp)

Version: 2
Date: 1 January 2015

Author's response to reviews:

We have clarified and revised the contributions of each author in the Authors' Contributions section. The study was non-randomised trial, so we cannot ensure CONSORT diagram.